Author's response to reviews

Title: Blood transfusion presenting with acute dyspnoea: a case report and review of literature

Authors:

SHEKHAR SHARMA (drshekharsharma@gmail.com)
ALTAF GAUHAR (altafhaji786@yahoo.com)
VIJAYKUMAR D K (dkvijaykumar@aims.amrita.edu)
JERRY PAUL (drjerrypaul@gmail.com)

Version: 6 Date: 4 August 2008

Author's response to reviews: see over
Point to point response to reviewers’ comments:

**Reviewer one:**

| Change in paragraph 4 to paragraph 2: | Complied with along with renumbering of the references. |

**Reviewer two:**

<table>
<thead>
<tr>
<th>Point number 1 (definition of ALI)</th>
<th>I have already referenced the said references, first one was deleted in favour of the second one (reference number 2 in the article) to comply with the limit of 15 references. The article discusses the definition of TRALI in box 1. However I am sorry for the non-inclusion of appropriate reference to box 1; this is now incorporated in the article.</th>
</tr>
</thead>
<tbody>
<tr>
<td>Point two (donor management)</td>
<td>I intended to discuss the clinical picture, management and investigations for TRALI in recipient. Donor management is a different topic requiring addition of a lot of text and references, and hence has been omitted in this article.</td>
</tr>
<tr>
<td>Point three (BNP)</td>
<td>the suggestions have been incorporated and role of BNP is now added to the discussion section. However this has raised the number of references from 15 to 16.</td>
</tr>
</tbody>
</table>